Author's response to reviews

Title: Matrix-metalloproteinases-1, -2, -3, -9, their inhibitors TIMP-1, -2, and the MMP1/TIMP1-complex in blood plasma as markers for transitional cell carcinoma of the bladder

Authors:

Andrea Staack (andrea.staack@charite.de)
Steffen Badendieck (badendieck@12move.de)
Dietmar Schnorr (dietmar.schnorr@charite.de)
Stefan Loening (stefan.loening@charite.de)
Klaus Jung (klaus.jung@charite.de)

Version: 2 Date: 22 November 2005

Author's response to reviews:

In my previous revision I have changed the initials of the authors and I have changed the email address of the second author.

Sincerely,

Andrea